Author's response to reviews

Title: Effects of type and level of training on variation in physician knowledge in the use and acquisition of blood cultures: a cross sectional survey

Authors:

Jorge P Parada (jorge.parada@med.va.gov)
David N Schwartz (david.schwartz@hektoen.org)
Gordon D Schiff (gdschiff@aol.com)
Kevin B Weiss (k-weiss@northwestern.edu)

Version: 7 Date: 8 September 2005

Author’s response to reviews:

See comments to the editor, as this revision only involves edits/formatting changes requested by the editor.